\begin{table}%t1
\par
\caption {\label{table1}Transiting Exoplanets characteristics. }
	%\centerline
 {
	\begin{tabular}{c  ccccccccccccc}		
	\hline\hline
& \multicolumn{7}{c}{Planet} & \multicolumn{6}{c}{Star}    \\
Object & $M$($\mjup$) & $R$($\rjup$) & Period & a(au) & log~($F_{\rm inc}$) & $e$ & $i$ & Spec. Type & $\te (K)$ & Age(Gyr) & $M$($\msun$) & [Fe/H] & $R$($R_{\odot}$) \\
\hline
CoRoT-1b$^a$ & $1.03\pm0.12$ & $1.49\pm0.08$ & 1.50 & 0.0254 & 9.467 & -- & 85.1 & G0V & 5950 & 0.2--4 & 0.95 & --0.3 & 1.11 \\
CoRoT-2b$^b$ & $3.31\pm0.16$ & $1.465\pm0.029$ & 1.74 & 0.0281 & 9.102 & -- & 87.84 & K0V & 5625 & 0.2--4 & $0.97\pm0.06$ & & 0.902 \\
CoRoT-3b$^c$ & $21.66\pm1.0$ & $1.01\pm0.07$ & 4.25 & 0.057  & 9.277 & -- & 85.9 & F3V & 6740 & 1.6--2.8 & 1.37 & --0.02 & 1.56 \\
CoRoT-4b$^d$ & $0.72\pm0.08$ & $1.19^{+0.06}_{-0.05}$ & 9.20 & 0.09 & 8.468 & -- & 90 & F0V & 6190 & $1.^{+1.0}_{-0.3}$ & 1.1 & & 1.15 \\
HAT-P-2b$^e$ & $8.04\pm 0.40$  & $0.98\pm0.04$  & 5.63 & 0.0677 & 8.962 & 0.5163 & 90 & F8  & 6290 &$2.7\pm1.4$ & 1.31 & 0.12 & 1.48 \\
\hline  
	\end{tabular}}
\par
	\smallskip
	 { $^a$ \cite{BBA08}, $^b$ \citet{AAB08}, $^c$ \citet{DDA08}, $^d$ \citet{MBG08}, $^e$ \citet{WJP07}.}\\
	 For all the CoRoT objects, $e=0$ is assumed in the analysis of the light-curve and of the radial velocity data.
	\end{table}